Author's response to reviews

Title: A simple hepatic cyst with elevated cyst fluid CA19-9 level: a case report

Authors:

Hidekatsu Yanai (yanaih@jikei.ac.jp)
Norio Tada (tada@jikei.ac.jp)

Version: 4 Date: 10 May 2008

Author's response to reviews: see over
Dear Sirs,

We will submit our manuscript entitled “A simple hepatic cyst with elevated cyst fluid CA19-9 level: a case report (657 words, 5 references, and 2 figures)” to *Journal of Medical Case Reports*.

We have no financial or other contractual agreements that might cause conflict of interest or be perceived as causing conflicts of interest. The corresponding author and all of the authors have read and approved the final submitted manuscript. No portion of the work has been or is currently under consideration for publication elsewhere. No portion of the manuscript other than the abstract has been published or posted on the internet. We will appreciate your consideration of our manuscript for publication in your journal.

Sincerely yours,

Hidekatsu Yanai, MD, PhD, FACP

Assistant Professor

Department of Internal Medicine,

The Jikei University School of Medicine,

The Member of American College of Physicians